Author's response to reviews

Title: Assessment of nerve involvement in the lumbar spine: association between magnetic resonance imaging, physical examination and pain drawing findings

Authors:

Bo C Bertilson (bo.bertilson@ki.se)
Eva Brosjö (evaxray@hotmail.com)
Hans Billing (hb@telemedicineclinic.com)
Lars-Erik Strender (Lars-Erik.Strender@ki.se)

Version: 6 Date: 13 July 2010

Author's response to reviews:

Information concerning informed consent has been added to the second paragraph of the method section.